Author's response to reviews

Title: Gallbladder adenocarcinoma with sarcoid-like reaction in regional lymph nodes: report of a case

Authors:

Yota Kawasaki (gekayota@gmail.com)
Kosei Maemura (kmaemura@hotmail.co.jp)
Hiroshi Kurahara (hiroshik45@gmail.com)
Yuko Mataki (mataki@m.kufm.kagoshima-u.ac.jp)
Satoshi Iino (iino@m2.kufm.kagoshima-u.ac.jp)
Masahiko Sakoda (sakoda@m.kufm.kagoshima-u.ac.jp)
Shinichi Ueno (ueno1@m.kufm.kagoshima-u.ac.jp)
Hiroyuki Shinchi (shinchi@m.kufm.kagoshima-u.ac.jp)
Sonshin Takao (sonshin@m2.kufm.kagoshima-u.ac.jp)
Shoji Natsugoe (natsugoe@m2.kufm.kagoshima-u.ac.jp)

Version: 4 Date: 23 November 2014

Author's response to reviews: see over
Dear Dr. Steve Povoski

We are very glad to have received a kind mail with comment for our manuscript entitled, “Gallbladder adenocarcinoma with sarcoid-like reaction in regional lymph nodes: report of a case” once again. According to the reviewers’ recommendation, the manuscript has been corrected and refined. The correction was indicated in red bold letters. We tried to answer reviewers’ comment in the author’s response sheet.

We would be grateful if this revised manuscript could be reviewed and considered for publication in *BMC Cancer*.

Please feel free to contact me if you have any questions or require further information.

Yours sincerely,

Yota Kawasaki

Kagoshima University Graduate School of Medical and Dental Sciences

Nov 23, 2014
Yota Kawasaki M.D
Department of Digestive Surgery
Breast and Thyroid Surgery Graduate School of Medical Sciences
Kagoshima University, Sakuragaoka 8-35-1, Kagoshima 890-8520, Japan.
Phone: 81-992-75-5361; Fax: 81-992-65-7426
E-mail: gekayota@gmail.com

Editor's Request:
"Please include individual author contributions for each author."
We have changed description to the followings.

>> YK performed the surgery, participated in the conception, coordinated with other doctors about the design of the study, collected data and a consent, carried out the histological evaluation and drafted the manuscript. KM participated in the surgery, histological evaluation and in drafting the manuscript. HK, YM and HS participated in the surgery and histological evaluation. SI, MS, SU, ST and SN conceived of the study, participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.